Author’s response to reviews

Title: Paranasal sinus air suction for the treatment of acute migraine – A randomized, double blind, pilot Study

Authors:

Rathnasiri bandara (rathnasirib68@gmail.com)
S samitha (samitha11@gmail.com)
AM Kiridana (kiridene@gmail.com)
udaya ralapanawa (ralapanawa@gmail.com)
tharuka Herath (tharukaherath111@gmail.com)

Version: 1 Date: 21 Sep 2018

Author’s response to reviews:

1. Figure:

We noticed that you have listed 1 figure legend in the manuscript. However, there is no figure uploaded in the system.

i uploaded the figure

2. We noticed that the order of the authors in the manuscript does not match to the order of the authors in the submission system. Please clarify which of the order is correct?

the one in the manuscript is correct